Author's response to reviews

Title: Analysis of undiagnosed tuberculosis-related deaths identified at post-mortem among HIV-infected patients in Russia: a descriptive study

Authors:

Yanina Balabanova (y.balabanova@qmul.ac.uk)
Vladimir Tchernyshev (orgmet-tptd@yandex.ru)
Igor Tsigankov (Glvrach-tptd@yandex.ru)
Svetlana Maximova (svetlana-maximova0@rambler.ru)
Natalya Mikheeva (natalyamikheeva@rambler.ru)
Ljudmila Fedukovitch (fedjuk11@rambler.ru)
Sergey Kuznetsov (noobesity@rambler.ru)
Ivan Fedorin (soptdlab@gmail.com)
Francis Drobniewski (f.drobniewski@qmul.ac.uk)

Version: 5 Date: 13 October 2011

Author's response to reviews:

The BMC Series Editorial Production Team 13 October 2011

Dear Production Team,

Thank you for your letter regarding the final proof of our manuscript “Analysis of undiagnosed tuberculosis-related deaths identified at post-mortem among HIV-infected patients in Russia: a descriptive study” accepted for publication by the BMC Infectious Diseases.

We have made all formatting changes that were required, proof-read the manuscript and hope that it is now ready for publication.

Sincerely,

Yanina Balabanova, Vladimir Tchernyshev, Igor Tsigankov, Svetlana Maximova, Natalya Mikheeva, Ljudmila Fedukovitch, Sergey Kuznetsov, Ivan Fedorin, Francis Drobniewski